

Journal of Quality Engineering and Production Optimization

Vol. 6, No. 1, Winter & Spring 2021 http://jqepo.shahed.ac.ir

Research Paper

DOI: 10.22070/JQEPO.2021.13366.1169

Order Picking Process: State-of-the-art on Classification

Amir Reza Ahmadi Keshavarz¹, Davood Jaafari^{1*}, Mehran Khalaj¹, Parshang Dokouhaki¹

¹ Department of Industrial Engineering, Parand Branch Islamic Azad University, Parand, Iran

* Corresponding Author: Davood Jaafari (Email:djaafari1350@gmail.com)

Abstract – An order-picking system (OPS) is essential to meet customer needs and orders in inbound logistics. Failure to perform the process properly results in high costs and customer dissatisfaction. To properly execute the OPS and to be able to meet the needs of customers with maximum efficiency, different aspects of the order picking process should be carefully considered. Research works have been conducted to improve OPS. This article attempted to review study components, study methods, and results in internal logistics while focusing on OPS. We also examined the composition of different aspects of OPS, including cost, equipment, warehousing, assignment, routing, batching, sequencing, and tardiness in 157 articles performed from 1958 to 2020 and studies on the process of time (Makespan). The reviewed studies were classified based on the simultaneous use of batching variables, assignment, routing, sequencing, and tardiness to clarify the relevance and application of routing, assignment, batching, sequencing, and tardiness aspects in the order picking process.

Keywords-Picking, Routing, Batching, Assignment, Tardiness.

I. INTRODUCTION

Manufacturing companies confront different challenges to fulfill customer orders and distinguish between them. Many manufacturing companies consider reducing and eliminating costs and improving productivity in their logistic system important improvement activities (De Koster et al., 2007). On the other hand, the warehouse is one of the most critical in-house logistics components playing a crucial role, among other activities, in the supply chain of any manufacturing company. To meet customer needs, the OPS, as part of the warehouse preparation process, produces lines to have a special place in the studies (De Koster et al., 2010). 60% of all productive activities and 55% of warehouse operating costs include order picking process De Koster et al. (2010); in each supply system, there are partswarehousing-related activities, such as receiving parts, warehousing, ordering, and shipping along product lines (Van Gils et al., 2018a). For reducing warehouse operating costs, in assembly lines, it is highly significant for the company to supply necessary materials within the factory to promote its Competitive power. In a factory (company), materials should be provided with high efficiency and flexibility. Components should also be provided at assembly stations to facilitate assembly operations (Dallari et al., 2009). The multiplicity of demands in an internal supply system reflects the intrinsic complexity of the systems (Elsayed & Lee, 1996). The complexity of the material and parts transportation system will lead to an increase in production simultaneously. Researchers, in recent years, have been increasingly motivated to develop OPS enhancement policies (Franzke et al., 2017). internal logistics aspects also need a great deal of attention due to the constraints and resources available to reduce costs and increase the flexibility of production

schedule changes to meet customer demands by expanding the ability of line feeding systems and offshore components. In this regard, using new methods of feeding materials and parts such as picking seem necessary (Battini et al., 2017). Picking refers to some of the factors, including moving, picking, deploying, and packing parts according to the amount of order required to meet the production schedule (Žulj et al., 2018). Receiving materials and parts from the warehouse, preparing them, and moving them along product lines are the most strenuous activities in an internal logistic system. According to studies, they account for 65% of total operating costs in the internal logistics process. The order pickings are based on the workforce and the flexibility of the workforce against changes due to existing constraints (Grosse et al., 2015). Timing and sequencing are two critical factors in picking operations (Lerher, 2018).

Other concepts used in the picking process are as follows:

travel distance, travel time, layout, channel configuration, dimension and size of the picking, order batching, ordering of picking parts, minimizing travel distance, maximizing occupational health, picking list (required orders), Stress, and Psychological Conditions (Grosse et al., 2015).

The process of OPS is performed in different ways, as discussed below:

- 1- to allocate the piece to the picker
- 2- Picker assignment for parts

In the part assignment method for picking parts, the picker is carried by a parts transporter or courier container, or he picks up the pieces himself. (Valle et al., 2017). In this method, the picking station is defined separately for each individual. The type of transfer operation is also proportional to the kind of concerned picking process. The part assignment method is also ergonomically beneficial to the operator (Battini et al., 2017). On the other hand, the method of assigning a piece refers to time, when each part has its picker in a specific place (Valle et al., 2017). Figure 1. shows the process map of the picking operation.



Figure 1. Process map of the picking operation

Figure 1 illustrates OPS. As seen, picking begins with receiving parts from the warehouse (Kim & Moon, 2003). There are two different ways of warehousing: batching and picking operations. If the received items do not meet the quality standards and documentation required for the factory, they will be returned to the manufacturer's dock. Picking operations are performed either manually or automated, using semi-robotic or robotic equipment. Each moving part to the picking area is also selected based on batching order and picking operations. Because warehouses provide busy services to customers, poor performance can lead to high costs (job demand) and customer dissatisfaction (Wruck, Vis, & Boter, 2016). Newmarket developments encompassing e-commerce, globalization, increasing customer expectations, and new regulations have intensified the competition between warehouses. It has forced warehouses to manage large numbers of small orders in limited opportunities (March et al., 2015). Managing order-picking operations and investigating the relationships between order-planning problems are two complex tasks to do by warehouse managers (Gu, Goetschalckx, & McGinnis, 2007). A successful e-commerce ordering process contains multiple different suppliers and programs that need to be quickly implemented. In this regard, completing orders is one of the most important fundamentals to succeed in the online business.

Moreover, to overcome shortcomings in the ordering process, several aspects should be carefully considered. The easier the process for the end customers is, the more satisfied they would be with you. Furthermore, to decrease the warehouse costs and applying new methods in the internal procurement system, some recent trends need to be more efficiently performed in the procurement industry. Previous studies have focused on order-picking and route selection and warehousing operations. However, the problems of picking operations are integrated. Marchet et al. (2015) provided a comprehensive empirical analysis of the choice of order-picking systems (e.g., focus and strategic decision making) (Van Gils et al., 2018a,b).

Van Gils et al. (2018a) focused on a combination of tactical and operational planning problems and how they could support new market developments. In this regard, some issues, such as ergonomics and order tardiness, have been rarely considered by the researchers. On this account, this paper investigated how warehouse managers can take advantage of combining several operational planning issues by reviewing previous articles. This paper sought to answer the following questions through reviewing the scientific literature related to combining order-picking problems, such as cost/equipment/warehousing/assignment, routing, batching, sequencing, and tardiness:

- 1. What problems have the related literature mentioned in order picking?
- 2. How have the order-picking problems been resolved and categorized in the reviewed articles?
- 3. What kind of warehouses have the reviewed studies mentioned?

Therefore, the purpose of this study was to investigate aspects of OPS.

The remainder of this paper is organized as follows: Section 2 describes OPS literature taxonomy. Section 3 discusses the research background. Section 4 describes the literature review methodology. Section 5 mentioned OPS problem planning. The managerial implications resulting from the literature overview are discussed in Section 6. Finally, Section 7 presents conclusions.

II. ORDER PICKING SYSTEM LITERATURE TAXONOMY

According to the objective of this paper, the OPS literature taxonomy surveys the scientific literature related to combining order picking problems to answer the following questions:

- 1. What problems have the related literature mentioned in order picking?
- 2. How have the order-picking problems been resolved and categorized in the reviewed articles?
- 3. What kind of warehouses have the reviewed studies mentioned?

This paper is organized as follows to answer the above questions; first, it considers the OPS definitions. Next, some previously conducted studies in OPS are discussed to provide a clear picture of prior research in this field. Then, the research questions and their objectives are elaborated in detail based on the research gap found in the literature. Afterward, the research method is presented. Subsequently, in the following steps of this research, discussions, and conclusions will be conducted. Figure 2 illustrates the classification proposed by the OPS literature.



Fig. 2: Proposed classification to review the OPS literature

III. ORDER PICKING SYSTEM RESEARCH BACKGROUND

During the past decades, awareness about OPS problems has been increasing worldwide thanks to industry and research. However, despite many efforts over the past few decades, OPS has not yet effectively matured. Therefore, it is essential to pay attention to warehousing activities, such as parts receiving, warehousing, orders, internal routing (Van Gils et al., 2018).

It is also crucial to consider the OPS classification developed by researchers to understand previously conducted research on OPS and the boundaries of OPS research. For instance, (Laporte, 1992) addressed routing and collection problems about the geographical distribution of warehouses. For this purpose, they classified a wide variety of variables. Rouwenhorst et al., 2000 proposed another classification dealing with problems related to warehouse control and specifically the warehouse design. Furthermore, Roodbergen & Vis (2009) developed another category through reviewing the literature on OPS. This classification concentrated on a wide range of problems, such as configuration, travel time estimation, storage, location, and sequencing. They also introduced some solutions for the mentioned problems.

Also, Park & Kim (2010) proposed the problems related to school bus routing. In this matter, some limitations, such as the maximum capacity of the bus, the maximum time to get on and off a bus, and the time window, were discussed. This class of surveyed problems has several subsets, including device transportation selection, routing, transport schedules, and completion time of the operation. Meanwhile, Van Gils et al. (2018) developed a classification for warehouse managers' benefit. They stated that since warehouses offer several services to customers, the poor performance of warehouses may result in high costs and customer dissatisfaction with demand. Van Gils et al. (2018) also declared that new market developments force warehouses to handle many orders within tight time windows. Accordingly, Van Gils et al. (2018) studied the related review and classification of the scientific literature with combinations of tactical and operational order picking planning problems.

Investigating a bunch of articles has also shown that the new policies on warehousing, ordering, routing, batching using fuzzy parameters as well as innovative algorithms have been developed for planning resources and optimizing performance by considering the number of the workforce (Kaur & Kumar, 2011; Gupta & Kumar, 2012; Bahrami et al., 2017; Kuo et al., 2016; Scholz et al., 2016; Jia & Heragu, 2009). On the other hand, most of the studies on OPS design with a combined program have been conducted to re-evaluate warehouse activities, routing problems, equipment, order selection in a warehouse, and warehouse design (see Table I) (Van Gils et al., 2018; Laporte, 1992; Park & Kim, 2010; Roodbergen & Vis, 2009; Rouwenhorst et al., 2000; Hansen & Mladenovi, 2001). Besides, some studies have been performed on the components of the order picking process (Scholz et al., 2017; Zhang et al., 2017; Chen et al., 2015).

References	Articles	Performance index
(Van Gils et al.2018; De Koster et al. 2007; Rouwenhorst et al, 2000; Roodbergen & Vis., 2009)	5	Design
(Hansen & Mladenovi, 2001)	2	Process changes
(Laporte., 1992; Park & Kim, 2010)	2	Routing

Table I. Areas of Study in the Review Articles

The advantage of the present study over the one conducted by Van Gils et al. (2018) is addressing the issues of order tardiness and warehouses. Van Gils et al. (2018) proposed, on a large scale, tactical and strategic decision-making issues. Likewise, the research method used in their paper was mathematical programming, analytical models, and simulations that are different from the classification of our paper. The current work was tried to consider the necessity of internal aspects of procurement, the limitations and available resources to reduce costs and increase adaptability to customer requirements by increasing productivity, and the use of OPS systems. Although in the review studies, the mentioned factors in the OPS system have been classified in terms of tactic, strategy, and design. But they have failed to present a proper classification, including performance measurement indicators, problem-solving methods, planning, and combination. Thus, this paper reviewed a combination of cost, equipment, warehousing, assignment, routing, batching, sequencing, and tardiness in the scientific literature on a combination of planning problems. To achieve the aim of this research, which is to classify the order picking process, first, planning problems and their combination should be

surveyed. After that, the performance measurement indicators and the chosen solution methods should be specified and investigated. The relationships between planning problems should be finally analyzed based on such indicators and solutions.

IV. METHODOLOGY OF REVIEWING OPS LITERATURE

This research work chose a three-step strategy to identify the most critical research papers, database selection, search, and sample modification. The literature-review method is depicted in Fig. 3.

A. Database selection

To track academic publications, several database engines can be utilized, such as Scopus, Web of Science, PubMed, and Google Scholar. Research conducted by Falagas et al. (2008) showed that the Scopus database is superior to other databases. However, in another study by Wang and Waltman (2016), journal classification of Web of Science is preferred. This paper similarly employed Scopus and Web of Science database; Google Scholar is also used as an assistant tool.



Fig. 3. OPS-literature-review method

B. Sample Search

Target articles of the present research were limited to those papers on planning and selection problems. Interestingly, this paper can also show the potential relationship between two or more planning problems. This study

reviewed the recent literature in which critical issues in the picking process have been highlighted. It also focused on planning problems, such as time performance or productivity metrics that can affect economic objectives, which are highly important in warehouse operations. For this purpose, "picking, routing, batching, assignment, tardiness" were searched as keywords in papers titles, abstracts, and keywords among articles and review papers from 1958 to 2020 (as exhibited in Fig. 4) in both Scopus and Web of Science. The initial search returned 281 articles. Fig. 4 shows that 97 articles have been published between 2010 and 2020, accounting for 62% of the total article. Those articles, which have been published in English language journals with an impact factor of at least 0.5 (According to the impact factor of 2017 by Thomson Reuters) and have a quality level of Q1 to Q4, were also considered. Books and conferences were excluded from the category.



Figure 4. Article Sharing Rate based on Number of publication Years

C. Sample selection

By applying abstract reading of 281 papers in the first step and subsequently, by the in-depth reading of the whole paper in circumstances in which it is impossible to identify the suitability of papers, the number of 157 papers with the most relevant content is selected. The results of this paper revealed that only 77 studies had been published in Elsevier, 21 studies in Taylor and Francis, and 16 studies in IIE Transactions. Table II classifies the articles based on their publication.

Table II. Number of articles	published in the study
------------------------------	------------------------

Electronic Publications	Number of articles
Elsevier	77
Taylor & Francis	21
IIE Transactions	16
Springer	12
Informs	10
Wiley Periodicals	6
Emerald	5

	•
Electronic Publications	Number of articles
Bundesvereinigung Loistik	3
IFAC	2
World Scientific Publishing Co	1
Lippincott	1
Bailiere Tindall	1
Institute of mathematical statistics	1
Indin society for development and environment	1

Continue Table II. Number of articles published in the study

To achieve the aim of the study, two issues were considered as follows: articles related to planning problems and articles exploring the interaction between planning problems in manual order selection systems. Based on Fig. 5, in the current research, the quality level of 157 articles was reviewed. As can be seen, 99, 28, 25, and 5 of the reviewed articles have journals of Q1, Q2, Q3, and Q4 quality, respectively. So, it is evident that most of the articles have been selected from high-quality journals. Sources of articles can be observed in Table III.



Figure 5. Number of Qualitative articles in journals

V. DISCUSSION

In this section, to help the managers decide more efficiently and overcome problems in the order picking process, the 157 chosen articles were classified and evaluated from three perspectives. Firstly, they were surveyed considering the problems of order-picking planning issues and the combination of problems to determine what essential issues are there in this area and which problems are simultaneously considered to optimize the order-picking performance. Such classification helps warehouse managers determine how they can solve a combination of issues related to order-picking planning so that they consider some fundamental issues in the order picking process. Secondly, all the articles were classified based on the performance measurements to analyze the relationship between planning problems and identify the performance indicators and their impact on the combination of problems. Thirdly, the articles were classified based on the methods used to solve the combination of order-picking planning problems to help the warehouse managers make better decisions.

Table III. Number of articles published in the study

References	<i>Qualitative</i> <i>Survey</i> <i>Journals</i>
Scholz et al., 2017; Van Gils et al., 2018a; Öztürko et al., 2019; Van Gils et al., 2018b; Cergibozan & Tasan., 2019; Kaur & Kumar., 2011; Jane & Laih., 2005; Gupta & Kumar, 2012; ; Letchford et al., 2013; ; Cortes et al., 2017; Muppani et al., 2008b; Öztürkog et al., 2014; Hwang et al., 2004; Geisser & Greenhouse., 1958; Brynzer & Johansson., 1996; Yung & Chung., 2004; Roodbergen et al., 2015; ; Zulj et al., 2018; Acimovic et al., 2014; Aghezzaf , 2007; Andriansyah et al., 2001; Battini et al. (2016); Battini et al., 2017b; Bouleimen & Lecocq, 2003, Bozer and Kile, 2008;Caron et al., 1998; ; Chabot et al., 2017; Chackelson et al., 2013; Chen et al., 2016; Chen.t.l et al., 2015; Chen.c.y et al., 2015; Dagenais et al., 2008; De Koster et al., 2010; De Koster et al., 2007; De Koster et al., 2012; Elbert et al., 2017; Elsayed et al., 1993; Geem., 2012; Giannikas et al., 2017; Glock & Grosse, 2012; Glock et al., 2016; Grosse et al., 2013; Grosse et al., 2016; Hsieh & Huang, 2011; Calzavara et al., 2017; Scholz et al., 2016; Luo et al., 2016; Jewkes et al., 2016; Hsieh & Huang, 2011; Calzavara et al., 2017; Scholz et al., 2016; Luo et al., 2016; Jewkes et al., 2005; Pohl et al, 2009; Kuo et al., 2016; Chun et al., 2016; Luo et al., 2016; Manzini et al., 2015; Marchet et al., 2002; Matusiak et al., 2014; Matusiak et al., 2017; Mowrey & Parikh., 2014; Muppani & Adil., 2008; ap & Wu, 2012; Pan et al., 2015; Parikh & Meller., 2008; Park & Kim, 2010; Petersen ; 2002, Petersen et al., 2001; Roodbergen & Vis., 2009; Rouwenhorst et al., 2000; Roy, 2016; Hansen & Miadenovi, 2001; Schleyer & Gue., 2017; Volgenant & Jonker., 1982; Won & Olafson., 2005; Zhang et al., 2017; Volgenant & Jonker., 1982; Won & Olafson., 2005; Zhang et al., 2017; Helsgaun., 2000; Ardjmand et al., 2020;Kübler et al., 2020;Cano et al., 2020; Valle et al., 2020; Li et al., 2017;	Q1
Ene et al., 2012; Bahrami et al., 2017; Teng & Hwang., 2017; Çeven et al, 2015; Clarke & Wright., 1964; Everett., 1963; Held & Karp., 1970; ; Ratliff & Rosenthal., 1983 ; Azadnia et al., 2013; Chien et al., 2010; Cordeau et al., 2007; Dallari et al., 2009; De Vries et al., 2016; Franzke et al., 2017; Grosse et al., 2013; Henn, 2015; Henn et al., 2013; Isler et al., 2016; ; Jia & Heragu,2009; Koch & Wäscher., 2016; ; Kulak et al., 2012; Manzini et al., 2007; ; Marras et al., 1993; Tappia et al., 2017; Hong, et al., 2015; Yu et al, 2015; Scholz et al., 2017; Pansart et al., 2017	Q2
Grosse et al., 2017; Battini et al., 2017, Akilbasha et al., 2018; Roodbergen et al., 2008; Berglund & Batta, 2012; Clark & Meller., 2013; Davarzani & Norrman., 2015; Elsayed & Lee, 1996; Gademann et al., 2001; Gademann & van de Velde., 2005; Garg, 1986; Gong & De Koster, 2011; Gong & De Koster, 2008; Gue & Meller, 2009; Hall, 1993; Hart et al., 1968; Bevan, 2015; Jarvis & McDowell., 1991; De koster & Van der Poort., 1998; Ma et al., 2009; 2010; Petersen et ; Ramtin & Pazour., 2015 ; Roy et al., 2012 ; Thomas & Meller, 2014; Park et al., 1999;	Q3
Moeller., 2011 ; Albareda et al ; Dekker et al, 2004; Pohl et al., 2009; Tang et al, 2011	Q4

A. Investigating the problems of order-picking planning and their combination

In this section, planning problems and their combinations were surveyed and, all the articles with at least two simultaneous planning issues were analyzed.

1) Study the warehouse layout

A warehouse comprises various areas and domains, including transportation operations, parts receiving area, parts warehousing area, and order picking area (Gu et al., 2007). Aisle is also referred to as the path between the rows of a cellar. Also, block layout is used refers to the contents of the repository by the segmentation method (Hong, 2014). Moreover, in the recent literature related to customized picking, three types of assignments have been introduced to determine the optimum anchors.

- 1. Assignment of storage and batching
- 2. Allocating storage and routing location
- 3. Wrapping and routing



Figure 6. Order Picking Area in Stock (Roodbergen, 2007)

Figure 6 determines that the effect of the warehouse position on the pick travel distance is less than 6% and, the cascade is 1% if there are 15 pickings or more per route. Some of the reviewed works have also introduced a model to minimizing travel distance in a fixed-structure warehouse (Roodbergen et al., 2008), (Petersen, 1997), (Hall, 1993), examined the effect of parts layout on order-picking efficiency by an analytical estimation of travel distance in single-block warehouses.

Auditors	Layout method	Number
Gademann et al., 2001; Hwang et al., 2004; Gademann and Velde, 2005; Won & Olafson., 2005 ; Schrotenboer et al., 2017; De Koster et al., 2008 ; Ho & Lin, 2017; Tsai et al., 2008; De Koster et al., 2010; Henn, 2012; Theys et al., 2010; Henn and Wascher, 2012; Hong et al., 2015; Azadnia et al., 2013; Henn et al., 2013; Chen et al., 2015; Chen et al., 2015; Glock et al., 2016 ; De Kostereng et al., 2015; Manzini et al., 2007; Henn, 2015; Isler et al., 2016; Cortes et al., 2017; Hong et al., 2016; Ratliff & Rosenthal., 1983 ; Gong & De Koster, 2008 ; Gue & Meller, 2009	Parallel aisle	27
Roodbergen & de Koster., 2001	Parallel aisle and a middle aisle	1
Valle et al., 2017; Scholz et al., 2017; Li et al., 2017	Parallel aisle with a middle aisle	3
Bozer and Kile, 2008	Parallel aisle & one main aisle	1
Chen et al., 2016 Ling & Huang, 2011;	Parallel aisle with one cross aisle	2
Kulak et al, 2012	Single and multiple cross aisles	1
Matusiak et al., 2014	Parallel aisle with 3 cross aisles	1
Pan et al., 2015	Pick-and-Pass system	1
Pansart et al., 2017	Parallel aisle with multiple middle aisle	1
Pohl et al, 2009	Fishbone Aisles	1
Valle et al, 2020	parallel aisles and two or more cross-aisles	1
	Total	40

According to Table IV, most of the studies in the warehouse were on parallel aisles and, other methods were less favored by the researchers.

Van Gils et al. (2018) and Scholz et al. (2017) stated that the process of picking and selecting pieces for picking is complete when the picker returns to the original location after selecting the last batch. In these papers, the authors used the mathematical model to solve small samples; In contrast, for larger samples, they used a variable neighborhood descent (VND) algorithm than by numerical experiments showed that the algorithm offers excellent solutions for path optimization (Theys et al., 2010).



Figure 7. Order picking process aspects (Van Gils et al., 2018)

(Van Gils et al., 2018) Figure 7 introduces items in an interconnected structure, such as storage location assignment, zone assignment of the warehouse, picking area, ordering, routing, labor assignment, labor force level, and labor assignment.

2) Study the Components of the Order Picking Process

Investigating the assignment and sequencing of the batch to multi-pickers concurrently with the optimal routing problem, checking the overall time of the custom multi-picker picking process, or applying a categorized warehousing method are generally done to maximize profit. In Figure 8, phases occurring in the study section on the components of the order picking process can be observed. As shown, it tries to optimize order assignment (sorting and routing of the shippers resulting in minimizing process time). Also, the workload balanced between the shippers is maintained. Also, from the start to the end of the picking process, this distance is kept to optimize the assignment.



Figure 8. Frequency of study areas examined in this study

Performance index	Articles
Cost	51
Equipment	23
Storage space	76
Assignment	39
Routing	112
Batching	36
Sequencing	23
Tardiness	10
Ergonomy	6

Table V. The Picking Process Components Study Section

Furthermore, considering cost(Table V), there were 51 articles (Hall, 1993; Öztürko et al., 2019; vanGils et at al., 2018; Ene et al., 2012; Grosse et al., 2013; Battini et al., 2016; Grosse et al., 2015; Battini et al 2017; Jane & Laih, 2005; Akilbasha et al., 2018; Letchford et al., 2013; Muppani et al., 2008; Çeven et al., 2015; Acimovic et al., 2014; Azadnia et al., 2013; Chabot et al., 2017; Chackelson et al., 2013; Chien et al., 2010; Chen et al., 2016; Cheng et al., 2015; Cordeau et al., 2007; Dagenais et al., 2008; Dekker et al., 2004; Elbert et al., 2017; Aghezzaf , 2007; Elsayed et al., 1993; Gademann et al., 2001; Gademanm & van de Velde, 2005; Glock et al., 2016; Gong & De Koster, 2008; Grosse & Glock, 2013; Grosse et al., 2017; Gue & Meller, 2009; Hart et al., 1968; Held & Karp, 1970; Bevan, 2015; Battini et al., 2017; Kuo et al., 2016; Muppani & Adil, 2008b; Petersen, 1999; Petersen, 2002; Parikh & Meller, 2008).

Most of the cost-related studies focused on changes in the warehouse system, reductions in routes, improvements of the assignment process, changes in the shape and size of warehouses, travel times, and changes in warehouse design processes.

This group of studies has provided methods for solving order-picking problems to help managers make better decisions. They have also considered some of the most critical aspects of order-picking for identifying scheduling problems and issues with different topics. Further, some research works have investigated order-picking other than resources (Scholz et al., 2017; van Gils., 2018; Henn & Wäscher, 2012; Azadnia et al., 2013; Chen et al., 2015; Elsayed et al., 1993; Elsayed & Lee, 1996; Gue & Meller, 2009; Henn, 2015; Schrotenboer et al., 2017; Tsai et al., 2008). One study on latency has simultaneously investigated five critical aspects of the order-picking process (assignment, routing, batching, sequencing, and tardiness) (Bouleimen & Lecocq, 2003). Four more studies (Sholz & Wäscher, 2017; Azadnia et al., 2013; Elsayed & Lee, 1996; Tsai et al., 2008) have also simultaneously studied routing, batching, sequencing, and tardiness. Table VI provides an overview of the combination of order picking scheduling problems about various aspects of its implementation. Five main aspects defined in the order picking process were included in 30 research on implementing the combined planning process problem. Fourteen studies did not study these five categories and solely focused on ergonomics, cost, equipment, and storage space. Therefore, we can state that only 9% of the reviewed studies did not focus on the five main categories of research, indicating the importance of assignment, routing, batching, sequencing, and tardiness in the order picking process (Öztürkog et al., 2014; Roodbergen et al., 2015; De Vries et al., 2016; Gong & De Koster, 2011; Glock & Grosse, 2012; Gu et al., 2007; Gu et al., 2010; Lee et al., 1996; Park et al., 1999; Pohl et al., 2009; Rouwenhorst et al., 2000; Hansen & Mladenovi, 2001; Tappia et al., 2017; Thomas & Meller, 2014). According to Table VI, the most significant combination of routing and batching studies with 21 studies and assignment and routing with 18 cases reflects that researchers have focused on routing.

Chackelson et al. (2013) used both simulation solutions to simultaneously examine picking processes to reduce operating costs and a hybrid algorithm to generate batches and assignment and sequencing customer needs. Several studies have studied scheduling problems together in a multi-block warehouse while combining batching and routing into a single scheme. After assigning the order to the packer, any order picker can begin the journey to the warehouse to select and pick. The orders should be prioritized according to the production plan and delivered to the pickings. In the meantime, the lost time is one of the most important factors to consider to prevent tardiness and increase productivity. For example, when a picker stops to prepare an order batch (Van Gils et al., 2018), it is one of the cases of tardiness in the order picking process.

The studies reviewed in this paper have addressed the reduction or elimination of tardiness through redesigning the picking process, batch organization, routing, prioritization, and picking activities. In Fig. 8, it can be seen that the most studied areas were routing and storage space since a total of 98 articles studied warehousing and design of a picking warehouse to reduce or, in other words, optimize routes.

Moreover, to find the most economical routes, warehouse layout, piecewise assignment, and parallel scheduling of equipment pickings have been studied. Travel routes in all logistic processes are mainly the most influential factors in terms of costing and timing, as shown in Table V.

This section deals with researchers with several studies associated with the topic of this paper. Five articles had been conducted by Henn as a lead author or contributor. Four studies by E.H. Grosse, Hong & Sholz, as a lead author or contributor. Also, from each of the following researchers, 2 studies were investigated (Ma et al., 2009; Muppani & Adil, 2008; Hwang et al., 2004; Ene & Öztürk, 2012; Calzavara et al., 2017; Aghezzaf, 2007; Berglund & Batta, 2012; Bozer & Kile, 2008; Caron et al., 1998; Yu et al., 2015; Gademann et al., 2001; Pan & Wu, 2012; Franzke et al., 2017; Garg, 1986; Gong & De Koster, 2011; Gu et al., 2007; Gue & Meller, 2009; Ling & Huang, 2011; Pohl et al., 2009; Manzini et al., 2007; Matusiak et al., 2014; Parikh & Meller, 2008; Roy et al., 2012).

In Scholz et al. (2018), mathematical programming modeling was considered in the above categories for the first time. This research, indeed, evaluated order batching and picker routing in OPS under the title of; "the benefits of integrated routing."

Another article Scholz & Wäscher. (2017) stated that the order batching problem involves grouping a set of customized picking orders to minimizing the entire picker journey so that the picking will be considered based on the calculation of the picking trip length. The strategy of this research was to simplify the order picking activity by simplifying the custom picker journey.

Auditors	Assignment	Routing	Batching	Sequencing	Tardiness	SUM
(Bouleimen & Lecocq., 2003 ; Kübler et al, 2020)	Х	Х	Х	Х		2
(Scholz et al, 2017; De Koster et al.2007; De Koster et al.,2012; Ling & Huang, 2011; Ene et al., 2012; Chun et al., 2016; Matusiak et al. 2017)	Х	Х	Х			7
Hwang et al., 2004; Clarke & Wright., 1964 ; Brynzer & Johansson., 1996; Caron et al., 1998; Chien et al., 2010; Dekker et al, 2004; Franzke et al., 2017; Glock et al., 2016; Grosse et al., 2015; Jewkes et al., 2004; Manzini et al., 2015; Matusiak et al. 2017 ; Pan & Wu, 2012; Chun et al, 2016 ; De Koster et al.2007; Ramtin & Pazour., 2015; Yu et al, 2015; Scholz et al. 2017;)	Х	Х				18
(Elsayed et al, 1993; Henn, 2015)	Х		Х	Х	Х	2

Table VI: Areas of Study in the Picking Process

Auditors	Assignment	Routing	Batching	Sequencing	Tardiness	SUM
(Azadnia et al., 2013; Cheng et al., 2015; Elsayed & Lee, 1996; Tsai et al, 2008;)		Х	X	Х	Х	4
(Schrotenboer et al., 2017; Gue & Meller, 2009)		Х			Х	2
Henn, 2015			X	Х	Х	1
(Tang et al, 2011; Chackelson et al, 2013; Zhang et al, 2017)	X		Х			3
Van Gils et al., 2018; ; Ene et al., 2012; ; Zulj et al., 2018; Albareda et al, 2009 ; Chen et al., 2015; De Koster et al, 2007 ;De Koster et al, 2012; Gademann & van de Velde., 2005 ; Gong & De Koster, 2008; Henn & Wäscher., 2012; Hong et al., 2016; Ling & Huang, 2011; Koch & Wäscher., 2016; Kulak et al., 2012; Chun et al, 2016 ; Matusiak et al., 2014; Matusiak et al., 2017; Schleyer & Gue., 2012 ; Valle et al., 2017; Valle et al, 2020)		Х	Х			21
(Moeller., 2011; Scholz et al, 2016; Ho & Lin, 2017; Caron et al, 1998; Theys et al, 2010; De Koster et al, 2010; Scholz et al, 2016).		Х		Х		6
(Parikh & Meller, 2008; Chackelson et al, 2013; Zhang et al, 2017)			Х	Х		3
(De Koster et al, 2010 ;Bouleimen & Lecocq., 2003 ; Cano et al, 2020)		Х	X	Х		2
(Chackelson et al, 2013; Zhang et al, 2017)	Х		Х	Х		2

Continue Table VI: Areas of Study in the Picking Process

Cano et al. (2020) introduced a mathematical programming model for the order batching and routing problem and a model for the batching, sequencing, and routing issue. For this purpose, they provided some formulas for the distance traveled and the pick time in the picking process in single-block and multi-block warehouses.

Moreover, Battini et al. (2017) focused on human communication and ergonomic parameters in OPS by noting that human operators manually perform most warehouse activities. This paper also emphasized that performance and cost are profoundly dependent on availability and productivity affected by operator fatigue.

A Pareto chart was also created to understand the effect of the assignment policy on the targeted activities to ultimately both provide a quantitative approach to integrate the rate of energy consumed at the estimated time for the entire OPS and introduce rest and unemployment time. On the other hand, Battini et al. (2016) proposed a new model for understanding the dual efforts of custom converters by highlighting the human capacities, the concept of time off, and unemployment. This article titled "Fatigue and Recovery" provided opportunities in an order-picking system. It is clear that the goal of OPS is not only to reduce costs but also to achieve better performance and overall benefits. Improving the performance of such a system can the advantage of human factors so that they receive more attention. In such a system, the operator preparation time is also considered to promote the picker efficiency and analyze the whole picking process.

Furthermore, various forms of assignment of the official and effective model in warehouse areas with comparative numerical capability have been developed. For example, in a study entitled "Maverick picking," Glock et al. (2016) argued that having sufficient curiosity for new issues such as modifying the workflows in customized picking is a significant challenge for most warehouse managers. The purpose of the above research was to describe the Maverick

picking process, its executive roots, and shape to demonstrate its potential effect on the order picking process. In Franzke et al. (2017), the author developed an integrated model to improve ergonomic and economic performance in custom piking with rotary pallets. This study entitled "Order picking Routing in Warehouse: A Comparison between Innovative and Optimal Solutions" was conducted in a warehouse whose components were located on pallets in both upper and lower rows.

Additionally (De Koster & Van der Poort, 1998) aimed to find optimal order picking routes for two types of conventional warehouses in which pickers could operate in a central warehouse. This would be important since order picking is a centralized work process. Taking into account some concepts such as (First-come, First-served) (FCFS), some scholars have considered combined orders about the route and the order of picking items (De Koster et al., 2007). In another review, Pohl et al. (2009) investigated the control and design of an order-picking warehouse. The authors stated that picking is a centralized and costly activity for most warehouses. It should be mentioned that 55 % of the total warehouse costs are related to the order picking process. (De Koster, 2010). It also declared that this share would be allocated to optimal interior design, warehouse assignment methods, routing methods, order batching, and zoning. In the present study, we considered travel time as a distance-dependent item and stated that travel time increased by distance; thus, travel distance could be a significant goal in designing and optimizing order picking warehouses. Additionally, in warehousing-related literature, two types of travel distances have been used; 1) the average pick travel distance (or average travel length and 2) overall travel distance (Roodbergen & de Koster, 2001). In (De Koster et al., 2012), as shown in Fig. 9, the researcher showed the distribution percentage of order picking time types.



Figure 9. Distribution Percentage of Order picker Time Types

In De Koster et al. (2012), research was performed by determining the number of domains in a travel picking system and simultaneously sequencing from different regions with a group of custom pickers.

Moreover, a study entitled "Custom Warehouse Coordination and Production Planning in a Cabinet Production Factory" proposed a hybrid integer planning model for warehouse order picking coordination and production planning. Simulated analysis and optimization have also been provided by Ant Colony to ultimately reduce the overall duration of picking operations by pickers and increase the capacity of the pickings concerning orders (Öztürko & Hoser, 2019).

Another study entitled "a discrete channel design model for an order picking warehouse" attempted to develop a model to reduce the travel distance for picking operations, and thereby lowering the cost of operations. To this end, the concept of warehouse design with large discrete channels was introduced. There is a linear transverse corridor called a tunnel. An efficient algorithm is used to calculate the tunnel (Van Gils et al., 2018). In this article, research was conducted on how to increase the effectiveness of order pickings using integrated warehousing, branding, regional picking, and routing policy decision-making: Delivering parts on time reduces search time and component selection,

also, in Ene & Öztürk. (2012), a study on the assignment of warehouse space and optimization of order pickings in the automotive industry, the assignment and order of the collection system was developed using mathematical modeling and random evolutionary optimization.

B. Performance measure

This section classified articles based on the performance measurement parameters to analyze the relationships between planning problems. Papers were also considered based on two tactical details and the details under consideration (Fig. 10).



Figure 10. Study areas of the investigated articles

1) TARDINESS

This section assessed order picking performance using the performance index for peak and tardiness times (i.e., the difference between order completion time and order delivery time).

Table VII is an overview of tardiness minimization in the order picking process considered in selecting the literature. Accordingly, only assignment, routing, batching, and sequencing could affect tardiness minimization of the assumed goal. Table VII also shows nine research works aimed at tardiness minimization, five of which simultaneously investigated routing, batching, and sequencing.

Moreover, Table VII shows the works, which had focused on routing, batching, sequencing to minimizing the overall tardiness. Based on this Table, it can be seen that six out of nine research works aimed at tardiness minimization, three of which simultaneously studied routing, batching, and sequencing.

Auditors	Assignment	Routing	Batching	Sequencing
(Sholz & Wäscher, 2017 ; Azadnia et al, 2013; Chen et al, 2015; Elsayed & Lee, 1996 ; Tsai et al, 2008)		Х	Х	Х
(Henn, 2015 ; Elsayed et al, 1993)	Х		Х	Х
(Henn et al., 2013)			Х	Х
(Roy, 2016)				Х

Table VII. Overview of the reviewed articles in minimizing Tardiness

Furthermore, in Scholz & Wäscher. (2017) stated that customer orders are due to the availability of a specific time to collect, prepare, and sequence the orders according to customer needs. Some works had also simultaneously studied the categories of assignment, batching, and sequencing. For example, Elsayed et al. (1993) considered the problems of sequencing and their effects on batching by considering components' assignment for the JIT technique in production control.

2) COST

The process of receiving materials and parts from the warehouse and preparing is the most costly activity in the intra-factory logistics system (Glock et al., 2016). Here, to decrease the warehousing costs, attention to past research reveals that order picking as one of the methods employed in an intra-factory logistics system contains more than 55% of the warehouse operating costs (Grosse et al., 2017).

Table VIII shows the focus of the studied articles on routing, batching, sequencing to minimizing the cost.

Auditors	Assignment	Routing	Batching	Sequencing	Tardiness	Others
(Letchford et al., 2013; Giannikas et al., 2017; Dallari et al., 2009; Cordeau & Laporte., 2007; Aghezzaf., 2007; Acimovic & Graves., 2014; Isler et al., 2016; Mowrey & Parikh., 2014; Neumann & Medbo., 2010; Petersen et al., 1999; Petersen., 2002)		Х				
(Everett., 1963)	Х					
(Pan & Wu., 2012)	X		Х			
(Gong & De Koster., 2008)		Х	Х			
(Scholz et al., 2016)		Х		Х		
(Schrotenboer et al., 2017)		Х			Х	
(Won & Olafson., 2005)			Х			
(Kaur & Kumar., 2011; Battini et al., 2016; Gupta & Kumar., 2012; Hart et al., 1968; Luo et al., 2016; Dagenais et al., 2008; Marche et al.,2012)						X

Table VIII. Overview of the reviewed articles in minimizing cost

3) PRODUCTIVITY

Twenty-four further studies focusing on the goal of maximization of productivity were reviewed. The reviewed articles have pointed out that:

- 1. Small size orders are performed with a great variety of parts.
- 2. Some orders will affect the efficiency of the order picking process due to either the seasonality of the order, the intention of moving the order forward, or various delivery locations based on point-of-sale information.

Table IX shows an overview of the studied articles regarding the maximum productivity of the order-picking process.

Auditors	Assignment	Routing	Batching	Sequencing	Tardiness	Others
(Bahrami et al, 2017 ; Hong et al, 2015 ; Yu et al, 2015)			Х			
(Hong., 2014 ; Lee & Geem., 2005 ; Pohl et al, 2009 ; Park & Kim., 2010 ; Schwerdfeger & Boysen., 2017)		Х		-		
(Pan et al., 2015)				Х		
(Grosse et al., 2015 ; Battini et al., 2017; De Vries et al., 2016; Gu et al., 2007; Kim & Moon., 2003; Lee et al., 1996; Cortes et al., 2017).						Х
(Valle et al, 2017)		Х	Х			
(Clarke & Wright., 1964; Glock et al. 2016; Pan et al. 2015)	Х	Х				
(Parikh & Meller., 2008)			Х	Х		
(Scholz & Wäscher., 2017 ; De koster et al. 2007)	X	Х	Х			
(Elsayed & Lee., 1996)		Х	Х	Х	Х	

Table IX. Overview of the reviewed articles in maximization of Productivity

As can be seen in Table 9, most studies have focused on routing; however, in Elsayed & Lee. (1996), simultaneous research was performed on batching, routing, sequencing, and tardiness. Two Papers have simultaneously studied categories of assignment, routing, and batching to maximize the productivity in OPS.

4) SPACE OPTIMIZATION

De Koster et al. (2007) believed that OPS is one of the most challenging and costly activities of warehouses, leading to customer dissatisfaction if not done correctly. In the area of order selection, the installation of OPS facilities and equipment has been considered. Table X Shows the studies performed to optimize the space available for OPS.

Kübler et al. (2020) mentioned three problems: order assignment, order batching, and picking routing in the warehouse for quick order selection. This paper solved the problem in combination using an innovative iterative method.

Auditors	Assignment	Routing	Batching	Sequencing	Tardiness	Others
(Van Gils et al., 2018)					Х	
(Geisser & Greenhouse., 1958)				Х		
(Roy et al, 2012; Berglund& Batta., 2012)						
(Jewkes et al, 2004; Manzini et al, 2015)	X	Х				
(Guo et al., 2007)		Х		Х		
(Gue & Meller., 2009)		Х			Х	

Table X. Overview of the reviewed articles on Space optimization

Auditors	Assignment	Routing	Batching	Sequencing	Tardiness	Others
(Caron et al, 1998; De Koster et al, 2007 ; Ene & Öztürk, 2012; Kübler et al, 2020)	X	Х	Х			
(Matusiak et al, 2014; Cano et al, 2020; Valle et al, 2020)		Х	Х	Х		
(Ratliff & Rosenthal., 1983; Roodbergen et al., 2015; Gong & De Koster., 2011; Geem., 2012; Marras et al., 1993; Muppani & Adil., 2008; Pohl et al., 2009; Rouwenhorst et al., 2000; Thomas & Meller., 2014; Andriansyah et al., 2001)						X

Continue Table X. Overview of the reviewed articles on Space optimization

There were also 25 studies in this field, with four articles on assignment, routing, and batching simultaneously.

5) MAKESPAN

To minimize the order picking makespan, it is necessary to solve order batching simultaneously, assignment and pickers routing.

Regarding makespan minimization, according to Table XI, it is clear that most of the works are concurrent studies so that five papers were on the simultaneous analysis of the assignment, routing, and batching, one on the simultaneous analysis of assignment, routing, batching, and sequencing, one on the concurrent analysis of routing, batching, sequencing, and tardiness, and one paper on the concurrent analysis of sequencing, routing, batching, and performance tardiness. According to Table XI, it is also evident that in all studies, routing has been one of the most significant topics for researchers. There were only 23 research studies on routing that we will discuss in more detail in Table XI.

Table XI. Overview of the reviewed articles on Mini	imization of Makespan
---	-----------------------

Auditors	Assignment	Routing	Batching	Sequencing	Tardiness	Others
(van Gils et al., 2018)	X	Х	Х		Х	
(Tsai et al., 2008)		Х	Х	Х	Х	
(Bouleimen & Lecocq., 2003)	Х	Х	Х	Х		
(Ene & Öztürk., 2012; DeKoster et al, 2012; Ling & Huang., 2011; Chun et al, 2016; Matusiak et al, 2017)	х	Х	Х			
(Chackelson et al., 2013)	Х		Х	Х		
(De Koster et al., 2007)		Х	Х	Х		
(Caron et al. 1998)	Х	Х		Х		
(Hwang & Lee., 2004; Brynzer & Johansson., 1996; Dekker et al., 2004; Franzke et al., 2017; Grosse et al., 2017; Ramtin & Pazour., 2015; Yu et al., 2015)	х	Х				
(Tang et al. 2011)	X		Х			

Auditors	Assignment	Routing	Batching	Sequencing	Tardiness	Others
	Assignment	Kouting	Datening	Sequencing	1 al unicss	Others
(Van Gils et al., 2018; Zulj et al., 2018; Albared et al., 2009; Cheng et al., 2015; Gademann & van de Velde., 2005; Henn & Wäscher., 2012; Hong et al., 2016; Koch & Wäscher., 2016; Kulak et al., 2012; Schleyer & Gue., 2012; Gademann et al., 2001; Scholz & Wäscher., 2017)		Х	Х			
(Moeller., 2011; Ho & Lin., 2017; Theys et al. 2010)		Х		Х		
(Parikh & Meller. 2008)			Х	Х		
(Glock & Grosse, 2012; Jane & Laih., 2005; Jarvis & McDowell., 1991; Kuo et al, 2016; Petersen et al, 2005)	Х					
(Öztürko & Hoser., 2019; Teng & Hwang., 2017; Akilbasha & atarajan., 2018; Cortés et al., 2017; Roodbergen et al., 2008; Chabot et al., 2017; Chen et al., 2015; Clark & Meller., 2013; Davarzani & Norrman., 2015; Hall., 1993; Held & Karp., 1970; Helsgaun., 2000; MaLaporte., 1992; Lerher et al., 2015; Leung et al., 2018; Lu et al., 2016; Manzini et al., 2007; Marchet et al., 2012; Petersen et al., 1997; Petersen & Aase ., 2004; Roodbergen & de Koster., 2001; Volgenant & Jonker., 1982; Tappia et al., 2017)		Х				
(Bozer & Kile., 2008; Cergibozan & Tasan., 2019)			Х			
(Jane & Laih., 2005; Öztürkog et al. 2014; Yung & Chung., 2004; Calzavara et al. 2017; Ma et al. 2009; Park et al. 1999)						Х

Continue Table XI. Overview of the reviewed articles on Minimization of Makespan

C. Measurement and Solution Methods

This section classified all the reviewed articles according to the solving methods used in the reviewed articles. Table XII shows the frequency of the solution methods used in the articles.

Table XII.	. Solving	methods	used	in	Studies	
------------	-----------	---------	------	----	---------	--

Solution methods	Articles	Percentage
Mathematical model	39	27.0%
Geometric matrix	3	2.1%
GA	9	6.4%
ACO	3	2.1%
VND	4	2.8%
VNS	7	5.0%
PSO	5	3.5%
Sequence optimization	1	0.7%

Solution methods	Articles	Percentage
Other algorithms	33	22.0%
Simulation	30	21.3%
Fuzzy logic	4	2.8%
Graph	6	4.3%

Continue Table XII. Solving methods used in Studies

According to Table XII, it can be seen that the mathematical modeling method has been the most commonly used solution in the research works, while the least used has been sequencing optimization. In this table, we have listed the use of statistical.

Aspects Routing Sequencing Tardiness Assignment Batching Solution method Mathematical model Х Х Х Х Х Geometric matrix Х Х Х GA Х Х Х Х ACO Х Х Х VND Х Х Х Х Х VNS Х Х Х Х Х PSO Х Х Х Х Х Sequence optimization Х Х Other algorithms Х Х Х Х Х Х Х Х Х Х Simulation Fuzzy logic Х Graph Х Х Statistical survey Х Х Х Х Other Methods Х Х Х Х Х

Table XIII. Solving Research Methods in Aspects of the Picking Process

Research works in the field of order picking, the use of statistical analysis methods, mathematical modeling, geometrical matrix, genetic algorithm, ant algorithm VND, VNS, PSO, and other algorithms, simulation, fuzzy, graph, and other methods. According to Tables XII and XII, it can be seen that most of the research works had focused on mathematical modeling with a 27% assignment percentage and other algorithms. Moreover, in the investigated research works, mathematical modeling, a variety of linear, nonlinear, integer, and dynamic models have been used. These models are, indeed, is a set of mathematical parameters and indicators related to the performance of process variables with several parameters. Besides, the simulation experiments for the behavioral analysis of processes and their numerical experiments were examined.

In this review article, papers on OPS published from 1958 to 2020 with scientific degrees shown in Table III and Fig. 5 were comprehensively studied. First, the articles, according to Table IV, examined the warehouses mentioned in the papers, and then it was found that parallel aisles with 27 were the most studied warehouses. Next, the picking order problems were investigated (Fig. 8 and Table V). Then, the combinations of picking order problems were classified in terms of minimizing tardiness (Table VII). Moreover, those articles on cost minimization were addressed (Table VIII). Later, studies on maximum productivity and storage space improvement were addressed (Table IX and X, respectively). Finally, the objectives of articles on minimizing makespan were provided (Table XI). In the following, the research methods used in the reviewed papers were presented in Table XII, and the methods used were classified according to the order-picking problems in Table XIII.

VI. Results

The results of this review study revealed the importance of integrating the problems related to multi-objective picking planning to effective management of order picking operations. The results indicated that the time horizon of decisions could significantly affect the proper approach to solving the hybrid planning problems. On the one hand, issues can be combined by analyzing the interactions between predetermined policies specific to each planning problem. On the other hand, two or more planning problems can be integrated and solved.

Fig. 2 and Table VI present an overview of the approach implemented in most articles intended to solve any combination of picking planning problems, as well as the number of articles that analyze each combination. Regarding the first research question, Figure 8 and Table V show that in the order picking area, a combination of planning problems, such as cost/equipment/warehousing/ assignment, routing, batching, sequencing, and tardiness, have been considered.

In this matter, it was found that the most studied issue (about 112 articles) had been related to routing; in comparison, tardiness had received the least attention with ten articles, indicating that researchers' tendency towards studying the issue of routing compared to other aspects of picking operations. Through investigating the order picking problems in the articles (based on Table VI), it was found that most of the combined studies had been on routing and batching with 21 works, followed by assignment and routing with 18 cases. This reflected the fact that the research base in the order picking area is routing. One of the crucial points in the picking process improvement is the integrated review of all aspects of the picking process. For example, there are two studies on the categories of assignment, classification, and sequencing (Chackelson, 2013; Zhang et al., 2017), and two studies on the categories of assignment, batching, sequencing, and tardiness (ELSAYED, 1993; Henn, 2015) (see Table VI). As can be observed in Table IX, most studies have focused on routing to increase productivity. However, Elsayed & Lee (1996) explored a concurrent survey on batching, routing, sequencing, and tardiness and two simultaneous studies on batching, assignment, and routing to maximize the productivity in OPS (Scholz & Wäscher., 2017; De Koster et al. 2007).

De Koster et al. (2007) believed that OPS is one of the most costly and challenging warehousing activities that could lead to customer dissatisfaction if it is not properly performed. Moreover, in terms of order picking, the installation of OPS facilities and equipment has been considered. On the other hand, different combinations of picking problems have been utilized to improve storage space (Table X) and minimize the makespan (Table XI). In addition, According to the fields of study, the articles reviewed in this work can be approached in terms of either tactic or component.

The reviewed literature observed three types of articles: Articles presenting new approaches and methods in the picking process, reviewer articles, and articles focusing on the components of the order-picking process. The classification of these studies helps warehouse managers differentiate between planning problems, become aware of the minimum combinations used, and detect which planning problems should be simultaneously considered.

The warehouse is one of the main internal logistic components in any manufacturing company and plays a critical

role in organizations' supply chains. Among the considered activities in the warehouse, the order picking process (production lines) has a special place in studies as a process of preparing parts in stock to satisfy customers' needs.

It should be mentioned that 60% of all activities and 55% of costs in the warehouse include the order picking process (De Koster, 2010). The parts warehouse activities, such as receiving parts, warehousing, order classification, and transportation along with production lines, are critical in any internal supply system (Van Gils, 2018). Considering the third research question of the present study, Fig. 6 shows that one of the main uses of a cross-dock warehouse is to route the picker in cross aisles. Therefore, it is necessary to assign routes to reduce picker travel in a warehouse and decrease the number of visits to the block by passing the aisle.

In the previous studies, a classification of the warehouses can be seen. In Table IV, it is obvious that parallel-aisle warehouses were the most studied warehouses. The classification of these studies helps warehouse managers differentiate between planning problems, determine the least-used combinations, and specify which planning problems should be simultaneously considered.

VII. CONCLUSION

Newmarket developments such as e-commerce, globalization, rising customer expectations, and new market regulations have forced warehouses to handle more orders in less time. In this matter, awareness of the impact of individual planning on picking planning for performance management and thus increasing customer service are required in overall performance. In this paper, studies, which have highlighted the picking process's main issues, were classified by reviewing the literature.

By focusing on planning problems that affect an economic goal, such as time-related performance or productivity indicators, problems are investigated as these goals are the most important warehouse operations. In this paper, regarding a combination of order-picking planning problems, articles reviewed by analyzing different tactical and operational planning problems simultaneously were classified to determine which planning problems are related, how various individual planning issues are related, and how can warehouse managers take advantage of the combination of multi-objective planning problems to cope with new market developments. The obtained results confirmed that all planning problems have to be incorporated due to different time horizons of decisions made one after another. At the operational level, there is still a need to integrate more planning problems and calculate the real-world features. Based on Fig. 8, planning problems are identified. In this figure, the classification of the issues surveyed indicates tardiness in the least number of articles.

For example, tardiness, which is critical to show commitments to the customers, has been studied in 10 cases in the picking process. The routing problem has been investigated in 112 cases, indicating a lack of attention to challenging programs of customers in terms of simultaneous tardiness and routing. The analyses also revealed some innovative methods had been proposed to obtain suitable solutions. As different industry managers look for ideal, simple, intuitive, and reliable solutions, cost reduction solutions are suggested in the order picking process. Furthermore, based on Table V, there are 112 studies on routing, 76 studies on storage space, 51 studies on cost, 36 studies on batching, 39 studies on assignment, 23 studies on sequencing, and ten studies on tardiness, indicating an insufficient balance between in decision-making variables about planning problems as well as a tendency towards more studies about routing and storage space at a higher speed than other aspects. Additionally, taking the 51 articles on cost minimization, it is clear that this issue has not been sufficiently surveyed.

It should be noted that cost reduction includes a wide range of issues, one of which is tardiness that can reduce costs. According to Table XII, it can be observed that the most common solving method used in the literature is the mathematical modeling method; in contrast, the least common one is the sequence optimization method.

Concerning the literature and methodologies of research works carried out in the area of order picking, it was found

that statistical analysis methods, mathematical modeling, geometric matrix, genetic algorithm, ant algorithm, Variable Neighborhood Descent (VND) algorithm, variable neighborhood search (VNS) algorithm, Particle swarm optimization (PSO) algorithm, other algorithms, simulation, fuzzy, graph, and other methods have been significantly employed. Base on Table XII, it can be observed that most of the research works have emphasized mathematical modeling with a 27% allocation percentage and other algorithms.

Furthermore, based on Tables XII and XIII, mathematical models have been mainly used for routing and assignment problems to explore the aspects of the picking process simultaneously. This reflects that the majority of researchers have tried to minimize distance in order to reduce cost and time, mainly because routing is metric and calculable. Moreover, at the operational level, there is still a need to integrate more planning problems and calculate the real-world features.

Despite the importance of human resources in the warehouse environment, a small number of articles have evaluated the labor planning problems, such as determining the daily labor force level, assigning the labor in order picking areas, and assigning jobs to pickers.

It should be further noted that as the warehouses offer multiple services to customers, the availability and performance of human resources ensure the quality of service to customers and the performance of order picking. For instance, determining the level of the workforce combined with the assignment of order pickers and the planning problem makes focusing on the number of pickers a challenging opportunity for future research. Previous studies have mainly focused on reducing order picking time. Working in an atmosphere with increasing pressure and limited time raises the likelihood of selection errors. In recent years, the number of articles analyzing multi-objective picking planning problems has risen dramatically. Nonetheless, currently, research works are mostly on storage location, order batching, and routing. At the tactical level of picking a problem, little attention has been paid to research. Previous studies not only have combined planning problems but have also simplified the order collection systems. The results developed in this paper indicated the need for more research on the congestion of warehouse aisles and the combination of picking operations problems in the future.

REFERENCES

- Acimovic, J. & Graves, S.C. (2014). Making better fulfillment decisions on the fly in an online retail environment. Manuf. Serv. Oper. Manag, Vol. 17, pp. 34–51.
- Aghezzaf, E.H. (2007). Production planning and warehouse management in supply networks with inter facility mold transfers. European Journal of Operational Research, Vol. 182, pp. 1122–1139.
- Akilbasha, P. & atarajan, P.G. (2018). An innovative exact method for solving fully interval integer transportation problems. Informatics in Medicine Unlocked, Vol. 11, pp. 95–99,
- Albareda, S. M., Ayuso, A.A., Molina, E. & De Blas, C.S. (2009). Variable neighborhood search for order batching in a warehouse. Asia Pac. J. Oper. Res, Vol. 26, pp. 655–683.
- Andriansyah, R., De Koning, W.W.H., Jordan, R.M.E., Etman, L.F.P. & Rooda, J.E. (2011). A process algebra based simulation model of a miniload–workstation order picking system. Computers in Industry, Vol. 62, No. 3, pp. 292–300.
- Ardjmand, E., Ghalehkhondabi, I., Young II, W.A., Sadeghi, A., Sinaki, R.Y., Weckman, G.R. (2020). A Hybrid Artificial Neural Network, Genetic Algorithm and Column Generation Heuristic for Minimizing Makespan in Manual Order Picking Operations, Expert Systems with Applications, Vol. 148.Ref 113566
- Azadnia, A. H., Taheri, S., Ghadimi, P., Mat Saman, M. Z. & Wong, K. Y. (2013). Order batching in warehouses by minimizing total tardiness: A hybrid approach of weighted association rule mining and genetic algorithms. The Scientific World Journal, Article ID246578.

- Bahrami, B., Aghezzafa, E.H. & Limere, V. (2017). Using simulation to analyze picker blocking in manual order picking systems. Procedia Manufacturing, Vol. 11, pp. 1798 1808.
- Battini, D., Calzavara, M., Persona A., Sgarbossa F. & Valentina, V. (2017). Fatigue and recovery: research opportunities in order picking systems. IFAC PapersOnLine, Vol. 50, No. 1, pp. 6882–6887.
- Battini, D., Calzavara, M., Persona, A., & Sgarbossa, F. (2017b). Additional effort estimation due to ergonomic conditions in order picking systems. International Journal of Production Research, Vol. 55, No. 10, pp. 2764–2774.
- Battini, D., Glock, C.H., Grosse, E. H., Persona, A. & Sgarbossa, F. (2016). Human energy expenditure in order picking storage assignment: A bi-objective method. Computers & Industrial Engineering, Vol. 94, 147–157.
- Berglund, P. & Batta, R. (2012). Optimal placement of warehouse cross-aisles in a picker-to-part warehouse with class-based storage. IIE Transactions, Vol. 44, No. 2, pp. 107–120.
- Bevan, S. (2015). Economic impact of musculoskeletal disorders (MSDs) on work in Europe. Best Practice & Research Clinical Rheumatology, Vol. 29, No. 3, pp. 356–373.
- Bouleimen, K. & Lecocq, H. (2003). A new efficient simulated annealing algorithm for the resource-constrained project scheduling problem and its multiple mode version. European Journal of Operational Research, Vol. 149, No. 2, pp. 268–281.
- Bozer, Y. A. & Kile, J. W. (2008). Order batching in walk-and-pick order picking systems. International Journal of Production Research, Vol. 46, No. 7, pp. 1887–1909.
- Brynzer, H. & Johansson, M. I. (1996). Storage location assignment: Using the product structure to reduce order-picking times. International Journal of Production Economics, Vol. 46–47, No. 1, pp. 595–603.
- Calzavara, M., Glock, C.H., Grosse, E.H., Persona, A. & Sgarbossa, F. (2017). Analysis of economic and ergonomic performance measures of different rack layouts in an order picking warehouse. Computers & Industrial Engineering, Vol. 111 (September), pp. 527–536.
- Cano J.A., & Correa-Espinal, A., & Montoya, R.A.G. (2020). Mathematical programming modeling for joint order batching, sequencing and picker routing problems in manual order picking systems. Journal of King Saud University-Engineering Sciences. 32, 219–228
- Caron, F., Marchet, G. & Perego, A. (1998). Routing policies and COI-based storage policies in picker-to-part systems. International Journal of Production Research, Vol. 36, No. 3, pp. 713–732.
- Cergibozan, Ç. & Tasan, A.S. (2019). Order batching operations: an overview of classification, solution techniques, and future.research. J. Intell. Manuf., 1–15.
- Çeven, E, & Gue, K.R. (2015). Optimal wave release times for order fulfillment systems with deadlines. Transportation Science, Vol. 51, No. 1, pp. 52–66.
- Chabot, T., Lahyani, R., Coelho, L.C. & Renaud, J. (2017). Order picking problems underweight, fragility and category constraints. International Journal of Production Research, Vol. 55, No. 21, pp. 6361–6379.
- Chackelson, C., Errasti, A., Ciprés, D. & Lahoz, F. (2013). Evaluating order picking performance trade-offs by configuring main operating strategies in a retail dis- tributor: A design of experiments approach. International Journal of Production Research, Vol. 51, No. 20, pp.6097–6109.

- Chen, F., Wang, H., Xie, Y. & Qi, C. (2016). An ACO-based online routing method for multiple order pickers with congestion consideration in warehouse. Journal of Intelligent Manufacturing, Vol. 27, No. 2, pp. 389–408.
- Chen, T.L., Cheng, C.Y., Chen, Y.Y. & Chan, L. K. (2015). An efficient hybrid algo rithm for integrated order batching, sequencing and routing problem. International Journal of Production Economics, Vol.160 pp.158–167.
- Cheng, C.Y., Chen, Y.Y., Chen, T.L. & Yoo, J.W., J. (2015). Using a hybrid approach based on the particle swarm optimization and ant colony optimization to solve a joint order batching and picker routing problem. International Journal of Production Economics, Vol. 170, pp. 805–814.
- Chien, C.M., Gong, Y., De Koster, R. & VanNunen, J. (2010). A flexible evaluative framework for order picking systems. Production & Operations Management, Vol.19, No.1, pp.70–82.
- Chun, C.-C., Kang, J.-R., Hou, C.-C. & Cheng, C. Y. (2016). Joint order batching and picker Manhattan routing problem. Computers & Industrial Engineering, Vol. 95, pp. 164–174.
- Clark, K.A. & Meller, R.D. (2013). Incorporating vertical travel into non-traditional cross aisles for unit-load warehouse designs. IIE Transactions, Vol. 45, No. 12, pp. 1322–1331.
- Clarke, G.U., & Wright, J.W. (1964). Scheduling of vehicles from a central depot to a number of delivery points. Operations Research, Vol. 12 No. 4, pp. 568–581.
- Cordeau, J. F. & Laporte, G. (2007). The dial-a-ride problem: Models and algorithms. Annals of Operations Research, Vol. 153, No. 1, pp. 29–46.
- Cortes, P., Gomez, M., Munuzuri, R.A. & Correa-Espinal, A. (2017). A tabu search approach to solving the picking routing problem for large-and medium-size distribution centres considering the availability of inventory and K heterogeneous material handling equipment. Appl. Soft Comput, Vol. 53, pp. 61–73.
- Dagenais, S., Caro, J. & Haldeman, S. (2008). A systematic review of low back pain cost of illness studies in the United States and internationally. The Spine Journal, Vol. 8 No. 1, pp. 8–20.
- Dallari, F., Marchet, G. & Melacini, M. (2009). Design of order picking system. The International Journal of Advanced Manufacturing Technology, Vol. 42, No. 1, pp. 1–12.
- Davarzani, H. & Norrman, A.J. (2015). Toward a relevant agenda for warehousing research: literature review and practitioners' input. Logistics Research, Vol. 8, No. 1, pp. 1–18.
- De Koster, R. & Van der Poort, E. (1998). Routing order pickers in a warehouse: A comparison between optimal and heuristic solutions. IIE Transactions, Vol. 30, No. 5, pp. 469–480.
- De Koster, R., Le Duc, T. & Roodbergen, K. J. (2007). Design and control of ware- house order picking: A literature review. European Journal of Operational Research, Vol. 182, No. 2, pp.481–501.
- De Koster, R., Le Duc, T. & Zaerpour, N. (2012). Determining the number of zones in a pick-and-sort order picking system. International Journal of Production Research, Vol. 50 No. 3, pp. 757–771.
- De Koster, R., VanderPoort, E. & Wolters, M. (2010). Efficient order batching methods in warehouses. International Journal of Production Research, Vol. 37, No. 7, pp.1479–1504.

- De Vries, J., de Koster, R. & Stam, D. (2016). Aligning order picking methods, incentive systems, and regulatory focus to increase performance. Production and Operations Management, Vol. 0, No. 0, pp. 1–14.
- Dekker, R., de Koster, R., Roodbergen, K. J. & vanKalleveen, (2004). Improving order-picking response time at Ankor's warehouse. Interfaces, Vol. 34, No. 4, pp. 303–313.
- Elbert, R.M., Franzke, T., Glock, C.H. & Grosse, E.H. (2017). The effects of human behavior on the efficiency of routing policies in order picking: the case of route deviations. Computers & Industrial Engineering. Ind. Eng. 111, 537–551.
- Elsayed, E. & Lee, M.K. (1996). Order processing in automated storage/retrieval systems with due dates. IIE Transactions, Vol. 28, pp. 567–577.
- Elsayed, E., Lee, M.K., Kim, S. & Scherer, E. (1993). Sequencing and batching procedures for minimizing earliness and tardiness penalty for order retrievals. Inter- national Journal of Production Research, Vol. 31, pp. 727–738
- Ene, S. & Öztürk, N. (2012). Storage location assignment and order picking opti- mization in the automotive industry. Int J Adv Manuf Technol, Vol. 60, pp.787–797.
- Everett III, H. (1963). Generalized Lagrange multiplier method for solving problems of optimum Assignment of resources. Oper. Res, Vol. 11, pp. 399–417.
- Falagas, M.E., Pitsouni, E.I., Malietzis, G.A., Pappas, G., 2008. Comparison of Pubmed, Scopus, web of science, and Google scholar: strengths and weaknesses. Faseb. J.22, 338e342.
- Franzke, T. Grosse, E. H., Glock, C. H. & Elbert, R. (2017). An investigation of the effects of storage assignment and picker routing on the occurrence of picker blocking in manual picker-to-parts warehouses. The International Journal of Logistics Management, Vol. 28, No. 3, pp. 841–863.
- Gademann, N. van den Berg, J. & vanderHoff, H. (2001). An order batching algorithm for wave picking in a parallel aisle warehouse. IIE Transactions, Vol. 33, pp. 385–398
- Gademann, N. & van de Velde, S. (2005). Order batching to minimize total travel time in a parallel-aisle warehouse. IIE Transactions, Vol. 37, No. 1, pp. 63–75.
- Garg, A. (1986). Biomechanical and ergonomic stresses in warehouse operations. IIE Transactions, Vol. 18, No. 3, pp. 246–250.
- Geem, Z. W. (2012). Effects of initial memory and identical harmony in global optimization using harmony search algorithm. Applied Mathematics and Computation, Vol. 218, No. 22, pp. 11337–11343.
- Geisser, S. & Greenhouse, S.W. (1958). An extension of box's results on the use of the distribution in multivariate analysis. Ann. Math. Stat, Vol. 29, No. 3, pp. 885–891.
- Giannikas, V., Lu, W., Robertson, B. & McFarlane, D. (2017). An interventionist strategy for warehouse order picking: Evidence from two case studies. International Journal of Production Economics, Vol. 189, pp. 63–76.
- Glock, C. H. & Grosse, E. H. (2012). Storage policies and order-picking strategies in ushaped order-picking systems with a movable base. International Journal of Production Research, Vol. 50, No. 16, pp. 4344–4357.
- Glock, C.H. Grosse, E.H., Elbert, R. M. & Franzke, T. (2016). Maverick picking: The impact of modifications in work schedules on manual order picking processes. International Journal of Production Research, Vol. 55, No. 21, pp. 6344–6360.

- Gong, Y. & De Koster, R. (2008). A polling-based dynamic order picking sys- tem for online retailers. IIE Transactions, Vol. 40, No. 11, pp.1070–1082.
- Gong, Y. & De Koster, R. (2011). A review on stochastic models and analy- sis of warehouse operations. Logistics Research, Vol. 3, No. 4, pp.191–205.
- Grosse, E. H., Glock, C. H. & Neumann, W.P. (2017). Human factors in order picking: A content analysis of the literature. International Journal of Production Re-search, Vol. 55, No. 5, pp. 1260–1276.
- Grosse, E. H., Glock, C. H., Jaber, M. Y. & Neumann, W. P. (2015). Incorporating human factors in order picking planning models: framework and research opportunities. International Journal of Production Research, Vol. 53, No. 3, pp. 695–717.
- Grosse, E.H. & Glock, C.H. (2013). An experimental investigation of learning effects in order picking systems. Journal of Manufacturing Technology Management, Vol. 24, No. 6, pp. 850–872.
- Grosse, E.H., Glock, C.H. & Jaber, M.Y. (2013). The effect of worker learning and forgetting on storage reassignment decisions in order picking systems. Comput- ers & Industrial Engineering, Vol. 66, No. 4, pp. 653–662.
- Gu, J., Goetschalckx, M. & McGinnis, L.F. (2007). Research on warehouse operation: A comprehensive review. European Journal of Operational Research, Vol. 177, No. 1, pp.1–21.
- Gu, J., Goetschalckx, M., & McGinnis, L. F. (2010). Research on warehouse design and performance evaluation: a comprehensive review. European Journal of Operational Research, Vol. 203, No. 3, pp. 539–549.
- Gue, K. R. & Meller, R.D. (2009). Aisle configurations for unit-load warehouses. IIE Transactions, Vol. 41, No. 3, pp. 171–182.
- Gupta, A. & Kumar, A. (2012). A new method for solving linear multi-objective transportation problems with fuzzy parameters. Applied Mathematical Modeling, Vol. 36, pp. 1421–1430.
- Hall, R.W. (1993). Distance approximations for routing manual pickers in a warehouse. IIE Transactions, Vol. 24, No. 4, pp. 76-87.
- Hansen, P. & Mladenovi ´c, N. (2001). Variable neighborhood search: Principles and applications. European Journal of Operational Research, Vol. 130, pp. 449–467.
- Hart, P., Nilsson, N. & Raphael, B. (1968). A formal basis for the heuristic determination of minimum cost paths. IEEE Transactions on Systems Science and Cybernetics, Vol. 4, No. 2, pp. 100–107.
- Held, M. & Karp, R.M. (1970). The traveling-salesman problem and minimum spanning trees. Operations Research, Vol. 18, No. 6, pp. 1138–1162.
- Helsgaun, K. (2000). An effective implementation of the Lin-Kernighan traveling salesman heuristic. European Journal of Operational Research, Vol. 126, No. 1, pp. 106–130.
- Henn, S. & Wäscher, G. (2012). Tabu search heuristics for the order batching prob- lem in manual order picking systems. European Journal of Operational Research, Vol. 222, No. 3, pp. 484–494.
- Henn, S. (2015). Order batching and sequencing for the minimization of the total tardiness in picker-to-part warehouses. Flexible Services and Manufacturing Journal, Vol. 27, No. 1, pp. 86–114.

- Henn, S., Koch, S., Gerking, H. & Wäscher, G. (2013). A U-shaped layout for manual order picking systems. Logistics Research, Vol. 6, No. 4, pp. 245–261.
- Ho, Y.C. & Lin, J. W. (2017). Improving order-picking performance by converting a sequential zone-picking line into a zone-picking network. Comput. Ind. Eng, Vol. 113, pp.241–255.
- Hong, S. & Kim, Y. (2016). A route-selecting order batching model with the S-shape routes in a parallel-aisle order picking system. European Journal of Operational Research, Vol. 257, No. 1, pp. 185–196.
- Hong, S. (2014). Two-worker blocking congestion model with walk speed m in a no-passing circular passage system. European Journal of Operational Research, Vol. 235, No. 3, pp. 687–696.
- Hong, S., Johnson, A. L. & Peters, B. A. (2015). Order batching in a bucket brigade order picking system considering picker blocking. Flexible Services and Manufacturing Journal, Vol. 28, No. 3, pp. 425–441.
- Hwang, H., Oh, Y. & Lee, Y. (2004). An evaluation of routing policies for order-picking operations in low-level picker-to-part system. Int. J. Prod. Res, Vol. 42, pp. 3873–3889.
- Isler, C., Righetto, G. & Morabito, R. (2016). Optimizing the order picking of a scholar and office supplies warehouse. Int. J. Adv. Manuf. Technol, Vol. 87, pp. 2327–2336.
- Jane, C.C. & Laih, Y.W. (2005). A clustering algorithm for item assignment in a synchronized zone order picking system. Eur. J. Oper. Res, Vol. 166, No. 2, pp. 489–496.
- Jarvis, J. M. & McDowell, E. D. (1991). Optimal product layout in an order picking warehouse. IIE Transactions, Vol. 23, No. 1, pp. 93–102.
- Jewkes, E., Lee, C. & Vickson, R. (2004). Product location, Assignment and server home base location for an order picking line with multiple servers. Computers & Operations Research, Vol. 31, No. 4, pp. 623–636.
- Jia, J. & Heragu, S.S. (2009). Solving semi-open queuing networks. Operations Research, Vol. 57, No. 2, pp. 391-401.
- Kaur, A. & Kumar, A. (2011). A new method for solving fuzzy transportation problems using ranking function. Applied Mathematical Modeling, Vol. 35, pp. 5652–5661.
- Kim, K.H. & Moon, K.C. (2003). Berth scheduling by simulated annealing. Transportation Research Part B: Methodological, Vol. 37, No. 6, pp. 541–560.
- Koch, S. & Wäscher, G. (2016). A grouping genetic algorithm for the order batching problem in distribution warehouses. J. Bus. Econ, Vol. 86, pp. 131–153.
- Kübler, P., & Glock, C.H., & Bauernhansl, T. (2020). A new iterative method for solving the joint dynamic storage location assignment, order batching and picker routing problem in manual picker-to-parts warehouses. Computers & Industrial Engineering.147.106645
- Kulak, O., Sahin, Y. & Taner, M. E. (2012). Joint order batching and picker routing in single and multiple-cross-aisle warehouses using cluster-based tabu search al- gorithms. Flexible Services and Manufacturing Journal, Vol. 24, No. 1, pp. 52–80.
- Kuo, R., Kuo, P., Chen, Y.R. & Zulvia, F.E. (2016). Application of metaheuristics-based clustering algorithm to item assignment in a synchronized zone order picking system. Appl. Soft Comput, Vol. 46, pp. 143–150.

- Laporte, G. (1992), The vehicle routing problem: An overview of exact and approximate algorithms. European Journal of Operational Research, Vol. 59, No. 3, pp. 345–358.
- Lee, K. S. & Geem, Z. W. (2005). A new meta-heuristic algorithm for continuous engineering optimization: harmony search theory and practice. Computer Methods in Applied Mechanics and Engineering, Vol. 194, No. 36, pp. 3902–3933.
- Lee, S.G., De Souza, R. & Ong, E.K. (1996). Simulation modeling of a narrow aisle automated storage and retrieval system (AS/RS) serviced by rail-guided vehicles. Computers in Industry, Vol. 30, No. 3, pp. 241–253.
- Lerher, T. (2015). Travel time model for double-deep shuttle-based storage and re- trieval systems. International Journal of Production Research, Vol. 54, No. 9, pp. 2519–2540.
- Letchford, A.N., Nasiri, S.D. & Theis, D.O. (2013). Compact formulations of the Steiner traveling salesman problem and related problems. European Journal of Operational Research, Vol. 228, pp. 83–92.
- Leung, K., Choy, K., Siu, P.K., Ho, G., Lam, H. & Lee, C.K. (2018). A B2C e-commerce intelligent system for re-engineering the eorder fulfilment process. Expert Syst. Appl, Vol. 91, pp. 386–401.
- Li, J., Huang, R., Dai, J.B., 2017. Joint optimization of order batching and picker routing in the online retailer's warehouse in China. Int. J. Prod. Res. 55, 447–461
- Ling, L.F. & Huang, Y. C. (2011). New batch construction heuristics to optimise the performance of order picking systems. International Journal of Production Economics, Vol. 131 No. 2, pp. 618–630.
- Lu, W., McFarlane, D., Giannikas, V. & Zhang, Q. (2016). An algorithm for dynamic order-picking in warehouse operations. Eur. J. Oper. Res, Vol. 248, pp. 107–122.
- Luo, S., Kong, X. & Tingting, N. (2016). Spline Based Survival Model for Credit Risk Modeling. European Journal of Operational Research',10.1016/j.ejor.2016.02.050
- Ma, L. Chablat, D., Bennis, F. & Zhang, W. (2009). A new simple dynamic muscle fatigue model and its validation. International Journal of Industrial Elsayed & Lee, Vol. 39, No. 1, pp. 211–220
- Manzini, R., Accorsi, R., Gamberi, M. & Penazzi, S. (2015). Modeling class-based storage assignment over life cycle picking patterns. Int. J. Prod. Econ, Vol. 170 (Part C), pp. 790–800.
- Manzini, R., Gamberi, M., Persona, A. & Regattieri, A. (2007). Design of a class based storage picker to product order picking system. The International Journal of Advanced Manufacturing Technology, Vol. 32, o. 7–8, pp. 811–821.
- Marchet, G. Melacini, M., Perotti, S. & Tappia, E. (2012). Analytical model to estimate performances of autonomous vehicle storage and retrieval systems for product totes. International Journal of Production Research, Vol. 50, No. 24, pp. 7134–7148.
- Marchet, G., Melacini, M., & Perotti, S. (2015). Investigating order picking system adoption: A case-study-based approach. International Journal of Logistics Re- search and Applications, 18 (1), 82–98. doi: 10.1080/13675567.2014.945400
- Marras, W.S. Lavendar, S.A., Leurgans, S.E., Rajulu, S.L., Allread, W.G. & Fathal- lah, M.S. (1993). The role of dynamic threedimensional trunk motion in occupationally-related low back disorders. The effects of workplace factors, trunk position, and trunk motion characteristics on risk of injury. Spine, Vol. 18, No. 5, pp. 617–628.

- Matusiak, M., De Koster, R. & Saarinen, J. (2017). Utilizing individual picker skills to improve order batching in a warehouse. European Journal of Operational Research, Vol. 263, No. 3, pp. 888–899.
- Matusiak, M., de Koster, R., Kroon, L. & Saarinen, J. (2014). A fast simulated annealing method for batching precedenceconstrained customer orders in a warehouse. European Journal of Operational Research, Vol. 236, No. 3, pp. 968–977.
- Moeller, K. (2011). Increasing warehouse order picking performance by sequence optimization. Procedia Social and Behavioral Sciences, Vol. 20, pp. 177–185,
- Mowrey, C.H. & Parikh, P.J. (2014). Mixed-width aisle configurations for order picking in distribution centers. European Journal of Operational Research, Vol. 232, No. 1, pp. 87–97.
- Muppani VR, & Adil, GK. (2008b). A Branch and Bound Algorithm for Class-Based Storage-Location Assignment. Eur J Oper Res, Vol. 189, pp. 492–507.
- Muppani, V.R. & Adil, G.K. (2008a). Efficient formation of storage classes for warehouse storage location assignment: A simulated annealing approach. Omega, Vol. 36, No. 4, pp. 609–618.
- Neumann, W.P. & Medbo, L. (2010). Ergonomic and technical aspects in the redesign of material supply systems: Big cases vs. narrow cases. International Journal of Industrial Ergonomics, Vol. 40, No. 5, pp. 541–548
- Öztürko, O.G. & Hoser, D. (2019). A discrete cross aisle design model for order-picking warehouses. European Journal of Operational Research, Vol. 275, pp. 411–430.
- Öztürkog'lu, O., Gue, K. R. & Meller, R.D. (2014). A constructive aisle design model for unitload warehouses with multiple picking and deposit points. European Journal of Operational Research, Vol. 236 No. 1, pp. 382–394
- Pan, J. C. H., Shih, P. H. & Wu, M. H. (2015). Order batching in a pick-and-pass warehousing system with group genetic algorithm. Omega, Vol. 57, pp. 238–248.
- Pan, J. C.H. & Wu, M. H. (2012). Throughput analysis for order picking system with multiple pickers and aisle congestion considerations. Computers & Operations Research, Vol. 39 No. 7, pp. 1661–1672.
- Pansart, L., Catusse, N., Cambazard, H., 2017. Exact Algorithms for the Picking Problem. arXiv preprint arXiv:1703.00699.
- Parikh, P. J. & Meller, R. D. (2008). Selecting between batch and zone order picking strategies in a distribution center. Transportation Research Part E: Logistics and Transportation Review, Vol. 44 No. 5, pp. 696–719.
- Park, B.C., Frazelle, E.H. & White, J.A. (1999). Buffer sizing models for end-of-aisle order picking systems. IIE Transactions, Vol. 31, No. 1, pp. 31–38
- Park, J. & Kim, B.I. (2010). The school bus routing problem. European Journal of Operational Research, Vol. 202, No. 2, pp. 311–319.
- Petersen, C. G. & Aase, G. (2004). A comparison of picking, storage, and routing policies in manual order picking. International Journal of Production Economics, Vol. 92, No. 1, pp. 11–19.
- Petersen, C.G. (1997). An evaluation of order picking routeing policies. International Journal of Operations & Production Management, Vol. 17, No. 11, pp. 1098–1111.

- Petersen, C.G. (1999). The impact of routing and storage policies on warehouse efficiency. International Journal of Operations and Production Management, Vol. 19, No. 10, pp. 1053–1064.
- Petersen, C.G. (2002). Considerations in order picking zone configuration. International Journal of Operations & Production Management, Vol. 22, No. 7, pp. 793–805.
- Petersen, C.G., Siu, C., & Heiser, D.R. (2005). Improving order picking performance utilizing slotting and golden zone storage. International Journal of Operations & Production Management, Vol. 25 No. 10, pp. 997–1012.
- Pohl, L., Meller, R. & Gue, K. (2009). Optimizing fishbone aisles for dual-command operations in a warehouse. Naval Research Logistics, Vol. 56, No. 5, pp. 389–403.
- Pohl, L., Meller, R. & Gue, K. (2009a). An analysis of dual-command operations in common warehouse designs. Transportation Research Part E: Logistics and Transportation Review, Vol. 45, No. 3, pp. 367–379.
- Ramtin, F. & Pazour, J.A. (2015). Product Assignment problem for an AS/RS with mul- tiple in-the-aisle pick positions. IIE Transactions, Vol. 47, No. 12, pp. 1379–1396
- Ratliff, H. D., & Rosenthal, A. S. (1983). Order-picking in a rectangular warehouse: A solvable case of the traveling salesman problem. Operations Research, Vol. 31, No. 3, pp. 507–521.
- Roodbergen, K. J. & de Koster, R. (2001). Routing order pickers in a warehouse with a middle aisle', European Journal of Operational Research, Vol. 133, No. 1, pp. 32–43.
- Roodbergen, K. J., Sharp, G. P. & Vis, I.F.A. (2008). Designing the layout structure of manual order picking areas in warehouses. IIE Transactions, Vol. 40, No. 11, pp. 1032–1045.
- Roodbergen, K. J., Vis, I. & Don Taylor Jr, G. (2015). Simultaneous determination of warehouse layout and control policies. International Journal of Production Re- search, Vol. 53 No. 11, pp. 3306–3326.
- Roodbergen, K.J. & Vis, I.F. (2009). A survey of literature on automated storage and retrieval systems. European Journal of Operational Research, Vol. 194, No. 2, pp. 343–362.
- Rouwenhorst, B., Reuter, B., Stockrahm, V., Van Houtum, G.J., Mantel, R.J., & Zijm, W.H.M. (2000). Warehouse design and control: Framework and literature review. European Journal of Operational Research, Vol. 122, No. 3, pp. 515–533.
- Roy, D. (2016). Semi-open queuing networks: A review of stochastic models, solution methods and new research areas. International Journal of Production Research, Vol. 54, No. 6, pp. 1735–1752.
- Roy, D., Krishnamurthy, A., Heragu, S.S. & Malmborg, C.J. (2012). Performance analysis and design tradeoffs in warehouses with autonomous vehicle technology. IIE Transactions, Vol. 44, No. 12, pp. 1045–1060.
- Schleyer, M. & Gue, K. (2012). Throughput time distribution analysis for a one-block warehouse. Transportation Research Part E: Logistics and Transportation Review, Vol. 48, No. 3, pp. 652–666.
- Scholz, A. & Wäscher, G. (2017). Order batching and picker routing in manual orderpicking systems: the benefits of integrated routing. Cent. Eur. J. Oper. Res, Vol. 25, pp. 491–520.
- Scholz, A., Henn, S., Stuhlmann, M. & Wascher, G. (2016). A new mathematical programming formulation for the single-picker routing problem. Eur. J. Oper. Res, Vol. 253, No. 1, pp. 68–84.

- Scholz, A., Schubert, D. & Wäscher, G. (2017). Order picking with multiple pickers and due dates simultaneous solution of order batching, batch assignment and sequencing, and picker routing problems. European Journal of Operational Research, Vol. 263, No. 2, pp. 461–478.
- Schrotenboer, A.H., Wruck, S., Roodbergen, K.J., Veenstra, M. & Dijkstra, A.S. (2017a). Order picker routing with product returns and interaction tardiness. Int. J. Prod. Res, Vol. 55, pp. 6394–6406.
- Schwerdfeger, S. & Boysen, N. (2017b). Order picking along a crane-supplied pick face: The SKU switching problem. European Journal of Operational Research, Vol. 260, No. 2, pp. 534–545.
- Tang, L., Wang, G., Liu, J. & Liu, J. (2011). A combination of Lagrangian relaxation and column generation for order batching in steelmaking and continuous-casting production. Naval Research Logistics. (NRL), Vol. 584, pp. 370–388.
- Tappia, E., Roy, D., De Koster, R. & Melacini, M. (2017). Modeling, analysis, and design insights for shuttle-based compact storage systems. Transportation Science, Vol. 51, No. 1, pp. 269–295.
- Teng, S.S. & Hwang, M.H. (2017). An efficient order-picking route planning based on a fuzzy set method with a multiple-aisle in a distribution center. Procedia Manufacturing, Vol. 11, pp. 1856 – 1862.
- Theys, C., Bräysy, O., Dullaert, W. & Raa, B. (2010). Using a TSP heuristic for routing order pickers in warehouses. European Journal of Operational Research, Vol. 200, No. 3, pp. 755–763.
- Thomas, L.M. & Meller, R.D. (2014). Analytical models for warehouse configuration. IIE Transactions, Vol. 46, No. 9, pp. 928–947.
- Tsai, C. Y., Liou, J. J. H. & Huang, T. M. (2008). Using a multiple-GA method to solve the batch picking problem: Considering travel distance and order due time. International Journal of Production Research, Vol. 46, No. 22, pp. 6533–6555.
- Valle, C. A., Beasley, J. E. & da Cunha, A. S. (2017). Optimally solving the joint order batching and picker routing problem. European Journal of Operational Research, Vol. 262, No. 3, pp. 817–834.
- Valle, C. A., Beasley, J. E. & da Cunha, A. S. (2020). O Order batching using an approximation for the distance travelled by pickers. *European Journal of Operational Research*, Vol. 284, Issue 2, pp. 460-484
- Van Gils, T., Ramaekers, K., Caris, A. & de Koster, R. B. (2018a). Designing efficient order picking systems by combining planning problems: State-of-the-art classi- fication and review. European Journal of Operational Research, Vol. 267, No. 1, pp. 1–15.
- Van Gils, T., Ramaekers, K., Braekers, K., Depaire, B. & Caris, A. (2018b). Increasing order picking efficiency by integrating storage, batching, zone picking, and routing policy decisions. International Journal of Production Economics, Vol. 197, pp. 243– 261.
- Volgenant, T. & Jonker, R. (1982). A branch and bound algorithm for the symmetric traveling salesman problem based on the 1-tree relaxation. European Journal of Operational Research, Vol. 9, No. 1, pp. 83–89.
- Wang, Q., Waltman, L., 2016. Large-scale analysis of the accuracy of the journal classification systems of Web of Science and Scopus. Journal of Informetrics 10, 347e364.
- Won, J. & Olafson, S. (2005). Joint order batching and order picking in warehouse operations. International Journal of Production Research, Vol. 43, No. 7, pp. 1427–1442.

- Yu, Y., De Koster, R. & Guo, X. (2015). Class-based storage with a finite number of items: Using more classes is not always better. Production and Operations Management, Vol. 24, No. 8, pp. 1235–1247.
- Yung, C.C. & Chung,Y.L. (2004). Machine scheduling with job delivery coordination. European Journal of Operational Research, Vol. 158, pp. 470–487 2004.
- Zhang, J., Wang, X., Chan, F. T. S. & Ruan, J. (2017). Online order batching and sequencing problem with multiple pickers: A hybrid rule-based algorithm. Applied Mathematical Modeling, Vol. 45, pp. 271–284.
- Zulj, I., Kramer, S. & Schneider, M. (2018). A hybrid of adaptive large neighborhood search and tabu search for the order batching problem. European Journal of Operational Research, Vol. 264, No. 2, pp. 653–664.